Author's response to reviews

Title: 2D power Doppler-3D Ultrasound imaging of a cesarean section dehiscence with utero-peritoneal fistula: a case report.

Authors:

Pedro Royo (proyo@alumni.unav.es)
Manuel García Manero (mgmanero@unav.es)
Begoña Olartecochea (bolarteco@unav.es)
Juan Luis Alcázar (jlalcazar@unav.es)

Version: 5 Date: 6 November 2008

Author's response to reviews:

To the Editorial team and paper Reviewers,
First of all I would like to thank you the effort and dedication to manage and review our work.

To reviewer 1.
Thank you very much for your comments.
The change of the paragraph of the objective and introduction has been amended in the text, the result is significantly better.

To reviewer 2.
Thank you very much for your comments.
1. The utero-peritoneal definition has been added, in the discussion section, in the context of a differential diagnosis of other types of uterine fistulas.
2. The reference reported has been added in the last paragraph of the discussion in regards to the clinical significance of uterine dehiscence in pregnancy.